Author's response to reviews

Title: Novel GATA4 promoter mutation associated with congenital heart disease in South India

Authors:

Saidulu Mattapally (mattapallysaidulu@gmail.com)
Sheikh Nizamuddin (snizam001@gmail.com)
Kona S Murthy (konasmurthy@gmail.com)
Kumarasamy Thangaraj (thangs@ccmb.res.in)
Sanjay K Banerjee (skbanerjee@thsti.res.in)

Version: 2 Date: 21 November 2014

Author's response to reviews: see over
To
Dr. Sergi Castellvi-Bel
Editor
BMC Medical Genetics

Date: 20-11-2014

Re: Resubmission of revised manuscript entitled “GATA4 mutations associated with congenital heart disease in South Indian patients (MS: 1704347284141499)”.

Dear Dr. Sergi Castellvi-Bel:
Thanks for giving us an opportunity to revise and further improve our manuscript according to the reviewer’s suggestions. Here, we would like to resubmit our revised manuscript entitled “GATA4 mutations associated with congenital heart disease in South Indian patients (MS: 1704347284141499)” for publication in your esteemed journal “BMC Medical Genetics”.

According to the editor and reviewer’s suggestions, we have revised the whole manuscript. You will find below details of our point-by-point responses to the reviewer’s comments. All new modifications and changes are marked red colour in the revised manuscript. Hope you will find it in order.

We think the manuscript has been greatly improved after revisions and we hope that you will now find it suitable for publication in your esteemed journal “BMC Medical Genetics”.

Sincerely yours,

Sanjay K Banerjee.

An autonomous institute of Dept. of Biotechnology, Ministry of Science & Technology, Government of India